%%#######################################################################
%##  CS 690A (Fall 2022), Assignment #2, Questions #1 and 2            ##
%##       Student Name: XXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX               ##
%##              MUN #: XXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX               ##
%########################################################################
\documentclass[12pt]{article}

\begin{document}

\begin{center}
\bf
Computer Science CS690A, Assignment \#2, Questions \#1 and 2 \\
Name: XXXXXXXXXXXXXXXXXX \\
MUN \#: XXXXXXXXXXXXXXXXXX \\
\end{center}

\begin{description}
\item[Question \#1]{

Give {\bf complete and correct} references for the following:

    \begin{description}
    \item[a)]{
    Marr, D. (1982) W.H. Freeman. [Book]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[b)]{
    Forisek, Fun with Algorithms, 214-227. [Conference paper]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[c)]{
    Argall, B.D. et al (2009), 57, 469-483. [Journal paper]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[d)]{
    Dufay and Latombe (1984) International Journal of Robotics Research. [Journal paper]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[e)]{
    Veale and Keane, ``The Competence of Sub-Optimal Theories of Structure
     Mapping on Hard Analogies.'' [Conference paper]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \end{description}

\noindent
Complete references must give full author names (last name and initials), paper and 
book titles, journal names, and conference names; where appriopriate, book series
and volume number, journal volume and number, publisher name, publisher address,
and page numbers must also be given.
}
\item[Question \#2]{

Give {\bf full and complete} references (at least the minimum requested number and
as many as you can find for bonus marks) for the following topics:

    \begin{description}
    \item[a)]{
    Algorithms for softweare system modularization [Minimum required number: 3]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[b)]{
     $NP$-hardness or completeness results for the {\sc Longest Common
      Subsequence} problem and various restricted cases. [Minimum required
      number: 3]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \item[c)]{
    Surveys and/or reviews of using robots for construction, repair, and/or
     maintenance [Minimum required number: 3]

\begin{verbatim}
Answer:

...
\end{verbatim}
    }
    \end{description}

\noindent
Complete references must give full author names (last name and initials), paper and 
book titles, journal names, and conference names; where appriopriate, book series
and volume number, journal volume and number, publisher name, publisher address,
and page numbers must also be given.
}
\end{description}

\end{document}
